Author’s response to reviews

Title: Interprofessional Education in Graduate Medical Education: Survey Study of Residency Program Directors

Authors:
Morhaf Al Achkar (alachkar@iupui.edu)
Matthew Hanauer (mhanauer@umail.iu.edu)
Chantel colavecchia (ccolavec@iu.edu)
Dean Seehusen (DSEEHUSEN@msn.com)

Version: 3 Date: 11 Dec 2017

Author’s response to reviews:

Dear editor,

We have made the requested changes.

Best regards,
The authors

1. Please include, after the title of the manuscript (and before the background) the full list of authors names, institution addresses and email addresses, in accordance with BMC Medical Education Submission Guidelines -https://bmcmededuc.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

Response: we made the changes.

2. In the author contributions section there is an author listed as DH (however no author on the author list has these initials) and the author Dean Seehusen (who would be represented as DS) is not present in the author contributions. Please clarify these to ensure all authors are represented in the authors contributions.

Response: we made the changes.
3. Please remove the use of contractions, such as 'don't' in the abstract, and replace with the full-length phrases, such as do not.

Response: we made the changes.